Author's response to reviews

Title: Human resource management in the Georgian National Immunization Program: a baseline assessment

Authors:

Laura C Esmail (laura.esmail@utoronto.ca)
Jillian Clare Cohen-Kohler (jillianclare.cohen@utoronto.ca)
Mamuka Djibuti (m.djibuti@curatio.com)

Version: 6 Date: 26 July 2007

Author's response to reviews:

We have made the following final changes to the document:

1. Highlighting/Tracking - all highlighting/tracking was removed from the manuscript.

2. We changed Author’s contributions -> Authors’ contributions and Methodology -> Methods

3. References - references were edited to contain all authors' names for #11 and #19 instead of "et al".